Author’s response to reviews

Title: Static balance and function in children with cerebral palsy submitted to neuromuscular block and neuromuscular electrical stimulation: Study protocol for prospective, randomized, controlled trial

Authors:

Soráia Kason (skazon@hotmail.com)
Luanda A C Grecco (luandacollange@hotmail.com)
Hugo Pasini (hugopneto@yahoo.com.br)
João C F Corrêa (jcorrea@uninove.br)
Thaluanna C L Christovão (thaluannaclc@hotmail.com)
Paulo Tarso (paulo.tarso@uninove.br)
Lilian C G Marson (odontogiannasi@uol.com.br)
Paulo R G Lucareli (paulolucareli@uninove.br)
Luiz V F Oliveira (oliveira.lvf@uninove.br)
Afonso S I Salgado (afonsoisalgado@yahoo.com.br)
Luciana M M Sampaio (lucianamalosa@uninove.br)
Claudia S Oliveira (csantos@uninove.br)

Version: 2 Date: 31 January 2012

Author’s response to reviews: see over
Covering letter


The authors state that the material is original, has not been published elsewhere, and is being submitted only to the BCM Pediatrics. The authors also accept full responsibility for the accuracy of all content, release and assign all rights for the publication of the study protocol, and state that there are no conflicts of interest related to the research or the manuscript. Authorship has been granted only to those individuals who have contributed substantially to the research or manuscript.

The study received approval from the ethics committee of the Universidade Nove de Julho (Sao Paulo, Brazil) under protocol number 200903/2008. The protocol for this study is registered with the Brazilian Registry of Clinical Trials - ReBEC RBR-5qzs8h. The study received funding from the CNPq for their development. The Integrated Movement Analysis Laboratory receives funding from the University in July of nine (Brazil) and research project was funded by the Brazilian agency Fostering National Council of Scientific and Technological Development.

Cordially,

Corresponding Author: Claudia Santos Oliveira

Mailing Address: R. Itapicuru 380 apt 111 - Perdizes-CEP-05006-000 - São Paulo, SP - Brazil Fax: 55 11 3868 1681